Author's response to reviews

Title: Strengthening primary health care teams with palliative care leaders: protocol for a cluster randomized clinical trial

Authors:

Joan Llobera (jillobera@ibsalut.caib.es)
Noemí Sansó (noemi.sanso@uib.es)
Amador Ruiz (aruiz@ibsalut.caib.es)
Merce Llagostera (mllagostera@ibsalut.caib.es)
Estefania Serratusell (estefania.serratusell@ssib.es)
Carlos Serrano (cserrano@ibsalut.caib.es)
María Luisa Martín Roselló (marisamartin@cudeca.org)
Enric Benito (benitoenric@gmail.com)
Eusebio J Castaño (ecastanyo@dgpif.caib.es)
Alfonso Leiva (aleiva@ibsalut.caib.es)

Version: 1 Date: 20 Jun 2017

Author’s response to reviews:

Dear Editor of the BMC Palliative Care Journal.

I am pleased to answer your request

1. Ethical and Funding Approval Documentation

I sent the approval of Ethical and Funding Approval Documentation and translated versions of the relevant parts to BMCSeriesEditorial@biomedcentral.com.

2. Funding

The study was externally funded and peer-review by the funding body.

3. Study Status
The study status of the study is ongoing and currently we are collecting data.

4. Related Articles

There is not any previous publications of this study

5. Declaration Section

Completed

yours sincerely

Noemi Sanso Martinez